Author's response to reviews

Title: National pattern of grain products consumption among Canadians in association with body weight status

Authors:

Hassan Vatanparast (vatan.h@usask.ca)
Susan Whiting (susan.whiting@usask.ca)
Alomgir Hossain (alomgir.hossain@usask.ca)
Naghmeh Mirhosseini (namirhossini@gmail.com)
Anwar Merchant (MERCHANT@mailbox.sc.edu)
Michael Szafron (Michael.Szafron@usask.ca)

Version: 3 Date: 15 Jun 2017

Author's response to reviews:

Dear Editor:

Thank you for your comments. We fixed the errors in the manuscript (highlighted). Following please find a point to point responses.

1. Line 280 - still includes "risk". Please go through the text to make sure this terminology is removed.

Response: We removed the term risk and revised the relevant sentences across the manuscript wherever we presented information about our analyses and the interpretation.

2. Line 285-289 - it is not clear what you are trying to say, please revise.

Responses: Sorry for the typos. We fixed them.